Author's response to reviews

Title: Detection of a divergent Parainfluenza 4 virus in an adult patient with influenza like illness using next-generation sequencing

Authors:

Seweryn Bialasiewicz (seweryn@uq.edu.au)
Jodie McVernon (j.mcvernon@unimelb.edu.au)
Terry Nolan (t.nolan@unimelb.edu.au)
Stephen B Lambert (sblambert@uq.edu.au)
Guoyan Zhao (GZhao@pathology.wustl.edu)
David Wang (davewang@borcim.wustl.edu)
Michael D Nissen (Michael_Nissen@health.qld.gov.au)
Theo P Sloots (t.sloots@uq.edu.au)

Version: 4
Date: 10 April 2014

Author's response to reviews:

Dear Sir/Madam

Please find attached a copy of the newest revision of the manuscript with changes addressing the reviewer’s comment (see summary below). Additionally, reference styles have been corrected to BMC standards.

Reviewer:

“I was wondering whether the citation on line 161 (ref. 1-6) needs to be adapted to (ref. 1-5, 8) as done >on line 182?”

The reviewer is both correct and keen-eyed. This citation error occurred during the last revision, and has now been corrected to (1-5,8), as per reviewer suggestion.

Regards,

-Seweryn Bialasiewicz